Author's response to reviews

Title: Insufficient early detection of peripheral neurovasculopathy and associated factors in rural diabetes residents of Taiwan: A cross-sectional study

Authors:

Chia-Mou Lee (goodideal@mail.mhchcm.edu.tw)
Chang-Cheng Chang (changcc@adm.cgmh.org.tw)
Chyong-Fang Chang (cfchang@gw.cgust.edu.tw)
Mei-Yen Chen (meiyen@gw.cgust.edu.tw)

Version: 3 Date: 5 November 2014

Author's response to reviews: see over
Response to editor’s comments – MS-2069695831208673

Many thanks for the helpful revision comments. We have followed the comments. Please see the changes in red font.

1. Give the full name and affiliation of the ethics committee that approved the project.
   Ans. Please see the changes in p7 with red font.

2. State whether the consent was verbal or written.
   Ans. Please see the changes in p7 with red font.

3. In the Competing Interest Section, change author(s) to authors.
   Ans. Please see the changes in p13 with red font.